Author’s response to reviews

Title: Risk factors for pulmonary tuberculosis in Croatia: a matched case-control study

Authors:

Anamarija Jurcev-Savicevic (anamarijajs@gmail.com)
Prof Rosanda Mulic (rmulic@hotmail.com)
Bozica Ban (bbozica@yahoo.com)
Prof Karlo Kozul (kozul.karlo@yahoo.com)
Ljiljana Bacun-Ivcek (bacunivcекlj@hotmail.com)
Jasna Valic (valicjasna@hotmail.com)
Gordana Popijac-Cesar (gordanapopijaccesar@yahoo.com)
Snjezana Marinovic-Dunatov (MarinovicDunatov.S@gmail.com)
Majda Gotovac (gotovac.majda@hotmail.com)
Aleksandar Simunovic (aleksandars07@hotmail.com)

Version: 2 Date: 27 April 2013

Author’s response to reviews:

Dear Editor,

the name of Ethics Committee was included as well as the statement that parental informed consent from those patients under the age of 18 was received. We are willing to make any additional changes if necessary.

On behalf of the authors,

Jurcev Savicevic Anamarija